Author's response to reviews

Title: The associations of leg lean mass with foot pain, posture and function in the Framingham Foot Study

Authors:

Robert R McLean (rmclean@hsl.harvard.edu)
Alyssa B Dufour (alyssadufour@hsl.harvard.edu)
Patricia P Katz (patti.katz@ucsf.edu)
Howard J Hillstrom (hillstromh@hss.edu)
Thomas J Hagedorn (tommyhagedorn@gmail.com)
Marian T Hannan (hannan@hsl.harvard.edu)

Version: 2 Date: 16 May 2014

Author's response to reviews:

Dear Sir/Madam:

The manuscript is now formatted according to journal specifications, including title page (i.e., author information), abstract and references. We also added authors' contributions and the competing interest statement.

thank you for your patience